Author's response to reviews

Title: Study protocol: Determinants of participation and quality of life of adolescents with cerebral palsy: a longitudinal study (SPARCLE2)

Authors:

   Allan F Colver (allan.colver@ncl.ac.uk)
   Heather O Dickinson (heather.dickinson@ncl.ac.uk)

Version: 2 Date: 26 April 2010

Author’s response to reviews:

Dear Sir

In response to your e-mail of 26 April

1 I have changed the abstract to remove all abbreviations
2 At the end of the article, after Authors’ Contributions, I have added a section called ethics appraisal in which I name the ethics committee which gave approval
3 I have made two very small changes expanding "para" to "paragraph" in the body of the text and correcting the details of one of the funding bodies

Yours sincerely

Allan Colver